

Author's response to reviews

Title: HVC1 ameliorates hyperlipidemia and inflammation in LDLR^{-/-} mice

Authors:

Se-Yun Cheon (chunsay1008@naver.com)

Kyung-Sook Chung (adella76@hanmail.net)

Kyung-Jin Lee (dostudy@naver.com)

Ho-Young Choi (hychoi@khu.ac.kr)

In-Hye Ham (sayxhd@naver.com)

Dong-Hoon Jung (brinicle87@naver.com)

Yun-Yeop Cha (omdcha@sangji.ac.kr)

Hyo-Jin An (hjan@sj.ac.kr)

Version: 2 **Date:** 23 Feb 2017

Author's response to reviews:

The authors' response letter has been included as a supplementary file